Author's response to reviews

Title: Fatty acid status and behavioural symptoms of Attention Deficit Hyperactivity Disorder in adolescents: A case-control study

Authors:

Ashley L Colter (acolter@romponline.com)
Caroline Cutler (ccutler@uoguelph.ca)
Kelly A Meckling (kmecklin@uoguelph.ca)

Version: 3 Date: 18 January 2008

Author's response to reviews: see over
January 18, 2008

Dear Editor
Nutrition Journal

Dear Sir:
As requested in your email of January 3, 2007, we have revised the manuscript MS 6450534801530550 entitled “Fatty acid status and behavioural symptoms of Attention Deficit Hyperactivity Disorder in adolescents”. Specifically we have made the requested typographical corrections to the abstract as requested by the editor. We hope that the manuscript will now be considered suitable for publication.

Kind regards
Kelly Meckling, PhD
Associate Professor